Author's response to reviews

Title: Evaluation of antiarthritic activity of Strychnos potatorum Linn seeds in Freund's adjuvant induced arthritic rat model

Authors:

Sanmugapriya Ekambaram (sanmug77@gmail.com)
Venkataraman Subramanian (svee1944@rediffmail.com)

Version: 9 Date: 20 August 2010

Author's response to reviews:

Major revisions:
Author list: the list of authors has been written according to the specified format
Conclusions - Conclusions part has been included after the Discussion.
Figures: fig 1 is given in the manuscript part itself. the legend is now corrected to fig 2 & 3
Tables: the tables are now included in the manuscript part itself after references

Minor revisions:
1. the title is changed to sentence case.